Author's response to reviews

Title: Fear of hypoglycaemia in parents of young children with type 1 diabetes: a systematic review

Authors:

  Katharine Barnard (k.barnard@soton.ac.uk)
  Sian Thomas (sian.thomas@btinternet.com)
  Pamela Royle (p.royle@abdn.ac.uk)
  Kathryn Noyes (Kathryn.Noyes@luht.scot.nhs.uk)
  Norman Waugh (n.r.waugh@abdn.ac.uk)

Version: 6 Date: 6 July 2010

Author's response to reviews:

Thank you for your email. Unfortunately I don't really understand what you are asking me to do, so I have tried to make the revisions requested but please do let me know if this was not what you wanted.

Regards

Katharine